Synthetic sling or artificial urinary sphincter for men with urodynamic stress incontinence after prostate surgery: the MASTER non-inferiority RCT

Lynda Constable,^{1*} Paul Abrams,² David Cooper,¹ Mary Kilonzo,³ Nikki Cotterill,⁴ Chris Harding,⁵ Marcus J Drake,⁶ Megan N Pardoe,² Alison McDonald,¹ Rebecca Smith,⁷ John Norrie,⁸ Kirsty McCormack,¹ Craig Ramsay,¹ Alan Uren,² Tony Mundy,⁹ Cathryn Glazener¹ and Graeme MacLennan¹

¹Health Services Research Unit, University of Aberdeen, Aberdeen, UK
²Bristol Urological Institute, Southmead Hospital, Bristol, UK
³Health Economics Research Unit, University of Aberdeen, Aberdeen, UK
⁴Faculty of Health and Applied Sciences (HAS), University of the West of England, Bristol, UK
⁵Department of Urology, Freeman Hospital, Newcastle upon Tyne, UK
⁶Bristol Urological Institute, University of Bristol, Bristol, UK
⁷Research and Innovation, Southmead Hospital, North Bristol NHS Trust, Bristol, UK
⁸Usher Institute, Centre of Population Health Sciences, University of Edinburgh, Edinburgh, UK
⁹Urology, University College Hospital, London, UK

Orology, Orliversity College Hospital, London,

*Corresponding author l.constable@abdn.ac.uk

Declared competing interests of authors: Paul Abrams reports consultancies with Astellas Pharma Inc. (Tokyo, Japan), Ipsen (Paris, France) and Recordati (Milan, Italy); reports being a lecturer for Astellas Pharma Inc., Pfizer Inc. (New York, NY, USA), Cipla (Mumbai, India), Ferring Pharmaceuticals (Saintt-Prex, Switzerland) and Sun Pharma (Mumbai, India); and is a triallist for Astellas Pharma Inc., outside the submitted work. Chris Harding reports personal fees from Astellas Pharma Inc., Medtronic (Dublin, Ireland) and Allergan (Dublin, Ireland); grants from Medtronic; personal fees from Teleflex Medical (Wayne, PA, USA) and GlaxoSmithKline (Brentford, UK); and research grants from the National Institute for Health and Care Research (NIHR) and The Urology Foundation (London, UK) outside the submitted work. Marcus J Drake reports personal fees from Astellas Pharma Inc. and Pfizer Inc., outside the submitted work and is a trustee of the International Continence Society (Bristol, UK). Nikki Cotterill reports grants from NIHR during the conduct of the study. Craig Ramsay reports membership of the NIHR Health Technology Assessment (HTA) General Board (2017-present). John Norrie reports grants from the University of Aberdeen (Aberdeen, UK) and the University of Edinburgh (Edinburgh, UK) during the conduct of the study and declares membership of the following NIHR/Medical Research Council boards: Efficacy and Mechanism Evaluation (EME) Funding Board (chairperson, 2019-present), Cardiopulmonary Resuscitation Decision-making Committee (2016),

HTA Commissioning Board (2010–16), HTA Commissioning Sub-Board (Expression of Interest) 2012–16, HTA Funding Boards Policy Group (2016), HTA General Board (2016–19), HTA Post-Board Funding Teleconference (2016–19), NIHR Clinical Trials Unit Standing Advisory Committee (2017–present), NIHR HTA and EME Editorial Board (2014–19) and Pre-exposure Prophylaxis Impact Review Panel (2017–present).

Disclaimer: This report contains transcripts of interviews conducted in the course of the research and contains language that may offend some readers.

Published August 2022 DOI: 10.3310/TBFZ0277

Plain English summary

MASTER non-inferiority RCT Health Technology Assessment 2022; Vol. 26: No. 36 DOI: 10.3310/TBFZ0277

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain English summary

Leakage of urine associated with physical exertion (e.g. sporting activities, sneezing or coughing) is common in men who have undergone prostate surgery, but it is difficult to improve. Many men still leak urine 12 months after their prostate surgery and may continue to wear protective pads or sheaths. The most common operation to improve incontinence is implantation of an artificial urinary sphincter. An artificial urinary sphincter is an inflatable cuff that is placed around the urethra, the tube that drains urine from the bladder. The cuff is inflated and compresses the urethra to prevent leaking. When the man needs to pass urine, he must deflate the cuff by squeezing a pump placed in his scrotum, which releases the compression on the urethra and allows the bladder to empty. Recently, a new device, the male sling (made from non-absorbable plastic mesh), has been developed. The sling, which is surgically inserted under the urethra, supports the bladder, but, in contrast to the artificial sphincter, it does not need to be deactivated by a pump and, therefore, the patient does not need to do anything to operate it. A sling is also easier for the surgeon to insert than a sphincter. However, in some men, the sling does not provide enough improvement in incontinence symptoms and another operation, to place an artificial urinary sphincter, is needed.

The aim of this study was to determine if the male sling was as effective as the artificial urinary sphincter in treating men with bothersome incontinence after prostate surgery. The study took the form of a randomised controlled trial (the gold standard and most reliable way to compare treatments) in which men were randomised (allocated at random to one of two groups using a computer) to either a male sling or an artificial urinary sphincter operation. We asked men how they got on in the first 2 years after their operation.

Regardless of which operation they had, incontinence and quality of life significantly improved and complications were rare. A small number of men did require another operation to improve their incontinence, and it was more likely that an artificial urinary sphincter was needed, rather than another sling operation, if a male sling was not successful. Satisfaction was high in both groups, but it was significantly higher in the artificial urinary sphincter group than in the male sling group. Those who received a male sling were less likely than those who received an artificial urinary sphincter to say that they would recommend their surgery to a friend.

Health Technology Assessment

ISSN 1366-5278 (Print)

ISSN 2046-4924 (Online)

Impact factor: 4.014

Launched in 1997, *Health Technology Assessment* (HTA) has an impact factor of 4.014 and is ranked 27th (out of 108 titles) in the 'Health Care Sciences & Services' category of the Clarivate 2021 Journal Citation Reports (Science Edition). It is also indexed by MEDLINE, CINAHL (EBSCO Information Services, Ipswich, MA, USA), Embase (Elsevier, Amsterdam, the Netherlands), NCBI Bookshelf, DOAJ, Europe PMC, the Cochrane Library (John Wiley & Sons, Inc., Hoboken, NJ, USA), INAHTA, the British Nursing Index (ProQuest LLC, Ann Arbor, MI, USA), Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and the Science Citation Index Expanded[™] (Clarivate[™], Philadelphia, PA, USA).

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full HTA archive is freely available to view online at www.journalslibrary.nihr.ac.uk/hta.

Criteria for inclusion in the Health Technology Assessment journal

Reports are published in *Health Technology Assessment* (HTA) if (1) they have resulted from work for the HTA programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

Reviews in *Health Technology Assessment* are termed 'systematic' when the account of the search appraisal and synthesis methods (to minimise biases and random errors) would, in theory, permit the replication of the review by others.

HTA programme

Health Technology Assessment (HTA) research is undertaken where some evidence already exists to show that a technology can be effective and this needs to be compared to the current standard intervention to see which works best. Research can evaluate any intervention used in the treatment, prevention or diagnosis of disease, provided the study outcomes lead to findings that have the potential to be of direct benefit to NHS patients. Technologies in this context mean any method used to promote health; prevent and treat disease; and improve rehabilitation or long-term care. They are not confined to new drugs and include any intervention used in the treatment, prevention or diagnosis of disease.

The journal is indexed in NHS Evidence via its abstracts included in MEDLINE and its Technology Assessment Reports inform National Institute for Health and Care Excellence (NICE) guidance. HTA research is also an important source of evidence for National Screening Committee (NSC) policy decisions.

This report

The research reported in this issue of the journal was funded by the HTA programme as project number 11/106/01. The contractual start date was in September 2013. The draft report began editorial review in November 2020 and was accepted for publication in September 2021. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The HTA editors and publisher have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the draft document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research funded by the National Institute for Health and Care Research (NIHR). The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the HTA programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the NHS, the NIHR, the NHR, the NHR, the HTA programme or the Department of Health and Social Care.

Copyright © 2022 Constable *et al.* This work was produced by Constable *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaption in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Prepress Projects Ltd, Perth, Scotland (www.prepress-projects.co.uk).

NIHR Journals Library Editor-in-Chief

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

NIHR Journals Library Editors

Professor John Powell Consultant Clinical Adviser, National Institute for Health and Care Excellence (NICE), UK, and Professor of Digital Health Care, Nuffield Department of Primary Care Health Sciences, University of Oxford, UK

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editor-in-Chief of HSDR, PGfAR, PHR journals

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board. Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Professor Ken Stein Professor of Public Health, University of Exeter Medical School, UK

Professor Jim Thornton Professor of Obstetrics and Gynaecology, Faculty of Medicine and Health Sciences, University of Nottingham, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk